Author's response to reviews

Title: A heart failure self-management program for patients of all literacy levels: A randomized, controlled trial [ISRCTN11535170]

Authors:

Darren A DeWalt (dewaltd@med.unc.edu)
Robert M Malone (rmalone@med.unc.edu)
Mary E Bryant (bbryant@med.unc.edu)
Margaret C Kosnar (margaret_corley@med.unc.edu)
Kelley E Corr (kelley_corr@med.unc.edu)
Russell L Rothman (russell.rothman@vanderbilt.edu)
Carla A Sueta (paris@med.unc.edu)
Michael P Pignone (pignone@med.unc.edu)

Version: 6 Date: 21 February 2006

Author's response to reviews: see over
February 20, 2006

BioMed Central Editorial Team
BioMed Central Health Services Research
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UK

Dear Editorial Team,

I am pleased that BMC Health Services Research has provisionally accepted the manuscript: A heart failure self-management program for patients of all literacy levels: A randomized, controlled trial [ISRCTN11535170]. We appreciate that Drs. McAlister and Doughty invested the time to evaluate our revisions and had few comments for further revisions. As suggested by Dr. McAlister, we have included the unadjusted results in the abstract. We made no other substantive changes. We did correct a few typographical errors and formatted the paper as recommended by BMC.

The BMC HSR review process has yielded a stronger manuscript. I look forward to working with you to reach final acceptance.

Sincerely,

Darren A. DeWalt, MD, MPH